Author’s response to reviews

Title: Infective myositis, an uncommon presentation of melioidosis: A case report and review of the literature

Authors:

Nuwan Jayawardena (nnsjmed@gmail.com)
Udaya Ralapanawa (udayapralapanawa@yahoo.com)
Prabhashini Kumarihamy (prabhashinkumarihamy@yahoo.com)
Thilak Jayalath (thilakj@gmail.com)
Shashi Prabhani Abeygunawardana (thahirss@yahoo.co.uk)
Nadisha Dissanayake (Nadishadissanayake@hotmail.com)
Priyantha Dissanayake (dmpkdissanayake@gmail.com)
Jeevani Udupihille (Jeevani_u@yahoo.co.uk)
Neelakanthi Ratnatunga (Neela72002@yahoo.com)
Chamara Dalugama (chamaradalugama@yahoo.com)

Version: 3 Date: 10 Nov 2019

Author’s response to reviews:

The Editor,

Journal of Medical Case Reports

Dear sir/madam.

Infective myositis, an uncommon presentation of Melioidosis: A case report and review of literature.
Thank you for considering our manuscripts. I have done all the requested changes and answered to all question for editor and reviewers’ comments. I have attached the clean version of revised manuscript. Details of the changers and answers to editor and reviewers’ comments are mention below in this letter.

I would be grateful to you if you could consider this article to publish in your journal. Please contact me if you need further correction and I am happy to reply them.
Best regards,

Prabhashini Kumarihamy.

In response to the reviewers’ comments, I have answered the reviewers’ questions and made the necessary changes in the manuscript as below. All the removed words and sentences are marked in red and cut off across. All the new added words and sentences are in red. We, authors would like to thank editors and reviewers for reviewing and commenting on our manuscript which helps us to improve the quality our article.

Reviewer #1: 1.
Page no 5, Line no 91 - ‘4’ was deleted keeping the word ‘four’
Page no 6, line no 115- the word hepatosplenomegally was corrected as hepatosplenomegaly.
Page no 7, line no 130- ‘computed topography’ was corrected as ‘computed tomography’
Line no 137- we haven’t mention about malaria antibody assay. Here, we have mentioned about ‘assay for melioidosis antibodies’
Page no 7, line no 143- ‘CRP level was 15 u/L’ was corrected as ‘CRP level was 15 mg/L’
Page 8, line 158-159- ‘Our patient had been treated for pneumonia requiring ICU care three weeks prior to admission to us could be the primary presentation of melioidosis.’ Was written into 2 sentences as ‘Our patient had been treated for pneumonia requiring ICU care three weeks prior to admission to us. This could be the primary presentation of melioidosis.’
Page 9, line no 179- ‘4’ was removed
Page 9, line 181- ‘Other diagnosis test are’ corrected as ‘Other tests help in diagnosis are’
Page 9, line 182- ‘Hemagglutination’ was corrected as ‘Haemagglutination’
Page 9, line 187- ‘milioidosis’ was corrected as ‘melioidosis’
Page 9, line 188- ‘guideline’ was corrected as ‘guidelines’
Page 11, line 218- computed topography’ was corrected as ‘computed tomography’
Page 11, Line 214-216 - List of abbreviations, corrected and now all started with abbreviation

Reviewer #2:

Not asked for correction